Author's response to reviews

Title: Long-term outcomes and risk factors of thyroid dysfunction during pegylated interferon and ribavirin treatment in patients with chronic hepatitis C infection in Taiwan

Authors:
Yu-Kang Chang (lilytang419@gmail.com)
Yuan-Tsung Tseng (2LH101@tmh.org.tw)
Kou-Huang Chen (a0939640405@gmail.com)
Kow-Tong Chen (ktchen@mail.ncku.edu.tw)

Version: 4 Date: 27 Mar 2019

Author's response to reviews:

The availability of data and material section and authors contribution section had been corrected as editor's comments.